Author's response to reviews

Title: Evaluation of a communication skills program for first-year medical students at the University of Toronto

Authors:

Solomon M Shapiro (solomon_shapiro@camh.net)
William J Lancee (william.lancee@utoronto.ca)
Christopher M Richards-Bentley
(Christopher.Richards-Bentley@medportal.ca)

Version: 9 Date: 27 January 2009

Author's response to reviews:

We have added the phrase "*Corresponding author" on the first page of the manuscript. No other changes have been made.